Author's response to reviews

Title: Early initiation of sexual activity: a risk factor for sexually transmitted diseases, HIV infection, and unwanted pregnancy among university students in China

Authors:

Qiaoqin Ma (qiaoqinma@yahoo.com.cn)
Masako Ono-Kihara (okmasako@pbh.med.kyoto-u.ac.jp)
Liming Cong (conglm@hotmail.com)
Guozhang Xu (xugzh@nbcdc.org.cn)
Xiaohong Pan (xhpan310009@yahoo.com.cn)
Saman Zamani (sam261@yahoo.com.jp)
Shahrzad Mortazavi Ravari (sh_mortazavi@pbh.med.kyoto-u.ac.jp)
Dandan Zhang (zhangdd@nbcdc.org.cn)
Takayuki Homma (honma-kyt@umin.ac.jp)
Masahiro Kihara (poghse@pbh.med.kyoto-u.ac.jp)

Version: 4 Date: 7 April 2009

Author's response to reviews:

We provide full details for all references in accordance with the format of BMC Public Health. As well, we made some minor revisions, like adding email addresses for all the authors on the title page, and change Aim -> Background, and remove Keywords, and change conflicts of Interest -> Competing interests.